Author's response to reviews

Title: Effectiveness of a Parenting Programme in a Public Health Setting. A Randomised Controlled Trial of the Positive Parenting Programme (Triple P) Level 3 versus Care as Usual provided by the Preventive Child Healthcare (PCH)

Authors:

Willem Spijkers (w.spijkers@med.umcg.nl)
Daniëlle EMC Jansen (d.e.m.c.jansen@med.umcg.nl)
Gea de Meer (g.demeer@ggdfryslan.nl)
Sijmen A Reijneveld (s.a.reijneveld@med.umcg.nl)

Version: 2 Date: 25 February 2010

Author's response to reviews: see over
Dear Editor, Dear Dr Norton,

In reply to your e-mail please find enclosed our revised manuscript, “Effectiveness of a Parenting Programme in a Public Health Setting. A Randomised Controlled Trial of the Positive Parenting Programme (Triple P) Level 3 versus Care as Usual provided by the Preventive Child Healthcare (PCH)”, to be considered for publication as a study protocol in BMC Public Health. We appreciate the opportunity to revise our manuscript.

Description of changes made:
(1) We documented, within the manuscript file, the specific name of the ethics committee which approved the study (see the ‘Design’ section): The Medical Ethics Committee of the University Medical Center of Groningen.

We hope that this revised version can contribute to the contents of BMC Public Health. We look forward to hearing from you.

With kind regards, also on behalf of the co-authors,

Edwin Spijkers

Address of correspondence: Edwin Spijkers, University Medical Center Groningen, Department of Health Sciences, Section of Social Medicine, Antonius Deusinglaan 1, 9713 AV Groningen, the Netherlands
E-mail: w.spijkers@med.umcg.nl
Phone: +31 50 3638369